Author’s response to reviews

Title: Male gender preference, female gender disadvantage as risk factors for psychological morbidity in Pakistani women of childbearing age - a life course perspective

Authors:

Farah Qadir (farah.qadir@fjwu.edu.pk)
Murad M Khan (murad.khan@aku.edu)
Girmay Medhin (gtmedhin@yahoo.com)
Martin J Prince (martin.prince@kcl.ac.uk)

Version: 4 Date: 14 September 2011

Author’s response to reviews:

As advised the section on ethical considerations has been added to the methods section of the manuscript.